Tue Cct 13 17:00: 09 GV 2020

Year: 20 Doy: 287

Observer: mso
WEATHER COMMENT: ntotter: Tue Oct 13 17:00: 15 GVI 2020

Tenp: 47.1f, Humidity: 9% Pressure: 28.65in, Wnd: 9nph from 151degs, Skies: Crrus, Cirrostratus and Al tocumnul us cl ouds
presest throughout the sky, but nmostly to the North, West and South. Heavy mist in the saddle up to the 8000 |evel. Road
were wet for nost of the drive up indicating it rained quite hard | ast night.
end

GENERAL COMMENT BY ntotter: Tue Oct 13 17:12: 37 GVI 2020

Opened wi ndows upstairs
___end___

GENERAL COMMENT BY ntotter: Tue Cct 13 17:12:47 GMI 2020

PM Bl ew of f Kcor Ol
__end____

GENERAL COMMENT BY ntotter: Tue Oct 13 17:14:46 GVI 2020

Opened done and started SHABAR.

end

GENERAL COMMENT BY nctotter: Tue Oct 13 20:18:25 GV 2020

The Spar had a problemthis morning with the Dec arm The Dec arm position had driven itself all the way against the NEG (
North) EOT sensor and seened to be bound. The commands to slew the travel in the opposite direction did not seemto affect
the position. | renoved the NEG EOR sensor, and gave the Dec a slew rate comand of -10.00 and the DEC arm went back tow
ard the Honme sensor and continued to drive until the Dec armtravel hit the POS EOR/ EOT sensor. | noved the NEG EOT sensor
further in and re-installed the NEG EOR sensor and positioned it so that the NEG EOR sensor triggers just a couple of se
conds before the NEG EOT sensor. | tried The Home command but it did not work and the Dec armtravel ed past the Hone senso
r without stopping. | had to manually slew to Home, and then hit the Home button so that it went to Standby.

**NOTE** When trying to find Home fromthe NEG EOR/ EOT status input -10.00 for a Slewrate until the Home GUJ illum nates,
then stop the slew and put in Standby. Wien trying to find Honme from POS EOR/ EOT status input 10.00 for a slew rate unti
Honme GUI illuninates, then stop the slew and put in Standby.

end

GENERAL COMMENT BY ntotter: Tue Oct 13 20:19: 08 GV 2020

Kcor runni ng.

___end

GENERAL COMMENT BY ntotter: Tue Cct 13 21:43:26 GVI 2020

Lots of aerosols visible in the Kcor image, but sky appears clear and blue. Spar appears to be working correctly with no p

robl ens apparent.

__end__

GPS COWMMENT by MLSO Tue Cct 13 21:45:09 GV 2020

Successfully | ogged in to system

Good di sk mount

GPS software runni ng

Last 5 GPS data files are:

/ mt/usb/dat aouti q_2020 282 _2145.bin 2147483647

/ mt/usb/dataouti q_2020 283 2145.bin 2147483647

/ mt/usb/dataouti q_2020_284_2145.bin 2147483647



/ mt/usb/dat aouti g 2020 285 2145.bin 2147483647

/ mt /usb/dat aouti g 2020 286 _2145.bin 2147483647
end

GENERAL COMMENT BY ntotter: Tue Oct 13 21:45:22 GV 2020

Did not run calibration this nmorning because of time |ost working on Spar problem
end

WEATHER COMMVENT: ntotter: Tue Oct 13 22:40: 50 GMI 2020

G rrocunul us and Al tocunul us clouds form ng above and to the upper West of the observatory. Pausing Kcor
end

GENERAL COMMENT BY ntotter: Tue Oct 13 22:45:29 GV 2020

O ographic clouds formng to the North and West of Maunal oa summt.

___end
WEATHER COMMVENT: ntotter: Tue Oct 13 23:46: 00 GVI 2020
G rrocumul us and Al tocunulus clouds still overhead and to the West. C ouds still covering the sumrit. Thick gray Orographi
¢ clouds have pushed up fromthe North and West and are now covering the sunmt.
end_

GENERAL COMVENT BY ntotter: Tue Cct 13 23:46: 24 GMI 2020
Cl osed dome shutter doors and w ndows.

end
WEATHER COVMVENT: ntotter: Wed COct 14 00: 38: 09 GMI 2020
Fog and light rain has begun on Maunal oa.

end
ONSI TE STAFF: berkey, ntotter



